Author’s response to reviews

Title: The healthcare costs of intoxicated patients who survive ICU admission are higher than non-intoxicated ICU patients: A retrospective study combining healthcare insurance data and data from a Dutch national quality registry

 Authors:

Ilse van Beusekom (i.vanbeusekom@amc.uva.nl)
Ferishta Bakhshi-Raiez (f.raiez@amc.uva.nl)
Nicolette de Keizer (N.F.Keizer@amc.uva.nl)
Dylan De Lange (D.W.deLange@umcutrecht.nl)

Version: 2 Date: 22 Dec 2018

Author’s response to reviews:

22nd December 2018

Dear editor,

Dear reviewer,

Thank you for giving us another opportunity to improve our manuscript. The comments were only minor and intended to improve the readability of the manuscript. We appreciate the effort by the editor and we have (literally) changed everything according to the editor's suggestions.

You will find two manuscript: (1) a manuscript in which all changes are marked in red and (2) a manuscript that is clean.

On behalf of my co-authors,

Kind regards,

Dylan de Lange, MD PhD